Author's response to reviews

Title: Measuring organisational and individual factors thought to influence the success of Quality Improvement in primary care: a systematic review of instruments

Authors:

Sue E Brennan (sue.brennan@monash.edu)
Marije Bosch (marije.bosch@monash.edu)
Heather Buchan (Heather.Buchan@health.gov.au)
Sally Green (sally.green@monash.edu)

Version: 5 Date: 13 October 2012

Author's response to reviews:

13 October 2012
Implementation Science Editorial Team
BioMed Central

Dear Implementation Science Editorial Team,

Re. MS: 8828426636528229 - Copy-edited version

Thank you for sending the copy-edited version of our manuscript.

We have checked the copy-edited version and uploaded a clean version of our manuscript.

We accepted all changes except for the removal of numbers in paragraph two on page eight and paragraph one on page twelve. These numbers refer to numbering in Figure 1 and we wish to retain them so that readers are able to cross reference to the relevant part of the figure.

There were some changes made in copy-editing that we were unsure were intentional. We draw these to your attention, but have not changed the manuscript from the version you sent.

• The list of tables on page 43 had been deleted.
• The styles specifying different levels of headings had been removed from some, but not all, sections of the manuscript. For example, the levels were retained in the abstract and background, but were missing from the methods and discussion sections (all headings formatted only as bold normal irrespective of level), and from parts of results section (e.g. from page 16 onwards. Again, all headings were formatted as bold normal, irrespective of level).

We have also uploaded the final version of the additional files as pdf files.
Thank you and best regards.

Sue Brennan